Author’s response to reviews

Title: Community Paramedic Point of Care Testing: Validity and Usability of Two Commercially Available Devices

Authors:

Ian Blanchard (ian.blanchard@ahs.ca)
Ryan Kozicky (ryan.kozicky@ahs.ca)
Dana Dalgarno (dana.dalgarno@ahs.ca)
Justin Simms (justin.simms4@gmail.com)
Stacy Goulder (stacy.goulder@ahs.ca)
Tyler Williamson (tyler.williamson@ucalgary.ca)
Susan Biesbroek (Susan.Biesbroek@ahs.ca)
Lenore Page (lenore.page@ahs.ca)
Karen Leaman (Karen.Leaman@cls.ab.ca)
Suzanne Snozyk (Suzanne.Snozyk@cls.ab.ca)
Lyle Redman (Lyle.Redman@cls.ab.ca)
Keith Spackman (Keith.Spackman@ahs.ca)
Christopher Doig (chip.doig@gmail.com)
Eddy Lang (eddy.lang@ahs.ca)
Gerald Lazarenko (gerald.lazarenko@ahs.ca)

Version: 2 Date: 22 Feb 2019

Author’s response to reviews:

1a. Please move your Declarations section to after your Abbreviations at the end of your manuscript.

Complete
1b. Rename "Introduction" into "Background".

Complete

1c. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of author JS does not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

Thank you for this comment. We have revised his contributions and have added further detail that more accurately reflect JS’s contribution to this manuscript. We believe that he does meet criteria for authorship, but understand why this comment has been made as we were too vague in his contributions on the 1st draft. We apologize for this.

We have also taken this opportunity to review the contributions of the other authors, and feel that authorship is warranted for all others on the list.

1d. Please move your Keywords section to below your abstract.

Complete

1e. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. (Fig 2)

We have reviewed all figures and tables and all appear to be cited at least once in the text.

1f. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Complete